Author’s response to reviews

Title: Adherence to long-term use of renin-angiotensin II-aldosterone system inhibitors in children with chronic kidney disease

Authors:
Chien-Ning Hsu (chien_ning_hsu@hotmail.com)
Shiou-Huei Huang (cnhsu10@gmail.com)
You-Lin Tain (tainyl@cgmh.org.tw)

Version: 1 Date: 31 Dec 2018

Author’s response to reviews:

Dear Dr Zalm,

Please find enclosed our revised manuscript BPED-D-18-00706 entitled “Adherence to long-term use of renin-angiotensin II-aldosterone system inhibitors in children with chronic kidney disease”. We appreciate the reviewers’ thoughtful comments and suggestions. Please find the pages attached with the revised manuscript are our point-by-point responses to each of the comments.

Thank you for these comments and additions to our paper. We believe that the revisions based on these comments have made this a better and clearer paper. We sincerely hope that in its present revised form, our manuscript will now meet with your approval for publication in the BMC Pediatrics.

Respectfully submitted,

You-Lin Tain, MD, PhD
Division of Pediatric Nephrology, Department of Pediatrics,
Kaohsiung Chang Gung Memorial Hospital and Chang Gung University, College of Medicine,
123 Dabi Road, Niausung, Kaohsiung 83301, Taiwan